Author's response to reviews

Title: Deltamethrin induced neurotoxicity and its attenuation by vitamin E in rats

Authors:

mona K Galal Dr (monagalal12@gmail.com)
Abdel Azim A Khalaf Prof (monakham@yahoo.com)
Hanan A Ogaly Dr (hananogaly@cu.edu.eg)
Marwa A Ibrahim Prof (marwa199@gmail.com)

Version: 3  Date: 30 June 2014

Author's response to reviews: see over
Cover letter

I have the pleasure to submit our manuscript under title "Deltamethrin induced neurotoxicity and its attenuation by vitamin E in rats" to your respectable journal.

Type of paper: research article.
Manuscript word count: 4860.
All authors listed in the table have read and approved the submitted form of this manuscript.

Authors Names/Affiliations:

<table>
<thead>
<tr>
<th>Name</th>
<th>Affiliation</th>
<th>Email</th>
</tr>
</thead>
<tbody>
<tr>
<td>Mona Khames Galal</td>
<td>Biochemistry department Faculty of Veterinary Medicine, Cairo University</td>
<td><a href="mailto:monagalal12@gmail.com">monagalal12@gmail.com</a></td>
</tr>
<tr>
<td>AbdelAzim A. Khalaf</td>
<td>Department of Forensic medicine and Toxicology, faculty of Veterinary medicine, Cairo University</td>
<td><a href="mailto:Marwa199@yahoo.com">Marwa199@yahoo.com</a></td>
</tr>
<tr>
<td>Hanan Ogaly</td>
<td>Biochemistry department Faculty of Veterinary Medicine, Cairo University</td>
<td><a href="mailto:hananogaly@cu.edu.eg">hananogaly@cu.edu.eg</a></td>
</tr>
<tr>
<td>Marwa Ibrahim</td>
<td>Biochemistry department Faculty of Veterinary Medicine, Cairo University</td>
<td><a href="mailto:Marwa199@gmail.com">Marwa199@gmail.com</a></td>
</tr>
</tbody>
</table>

Corresponding Author
Name: Mona Khames Galal
Affiliation: Biochemistry department Faculty of Veterinary Medicine
Address: Faculty of Veterinary Medicine. Cairo University
Email: monagalal12@gmail.com
Fax: ±202-35725240

Conflict of interest/sponsorship (if any)
we have no conflict of interest

Ethical committee approved? YES NO